Author's response to reviews

**Title:** Solid variant of aneurysmal bone cyst of the heel; a case report

**Authors:**

Joanna Lekka (Jlekka@yahoo.gr)  
Theofani Gavressea (fangav@261.gr)  
Gabriella Stanc-Giannakopoulos (gabidr1999@yahoo.com)  
Nikolaos Demertzis (NDemertzis@yahoo.gr)

**Version:** 2  **Date:** 24 March 2010

**Author's response to reviews:**

The revised version of our case report has a title page with all the needed informations, we restructure the abstract so it would have three sections (introduction, case presentation and conclusion), and in the manuscript we included the ethnicity of the patient, a conclusion section, the confirmation of the written informed consent of the patient and information about the contribution of each author.